Author's response to reviews

Title: Survey and Analysis of the Current State of Residency Training in Medical School-Affiliated Hospitals in China

Authors:

Hong Bo (Bohong247@163.com)
Dong-Hua Zhang (donghuazhang@sohu.com)
Tian-Ming Zuo (zuotianmingming@qq.com)
Dong-Bo Xue (xue9971@sina.com)
Jin-Song Guo (guojinsong1234@sina.com)
Mei-Na Liu (liumeina1965@qq.com)
Jing-Zhu Dong (dongjingzhu1980@126.com)
Baozhi Sun (baozhisun@126.com)
Jin Zhou (zhoujin1111@126.com)

Version: 2
Date: 25 November 2013

Author's response to reviews:

1. We have added the email addresses of all authors in the title page.

2. We have added the ethics statement in Page 6 (at the end of part “Survey period and subjects”)